Author's response to reviews

Title: Pit excision with phenolisation of the sinus tract vs. radical excision in sacrococcygeal pilonidal sinus disease: study protocol for a single centre randomized controlled trial

Authors:

Edgar J.B. Furnee (ejbfurnee@outlook.com)
Paul H.P. Davids (pdavids@diakhuis.nl)
Apollo Pronk (apronk@diakhuis.nl)
Niels Smakman (nsmakman@diakhuis.nl)

Version: 5
Date: 11 February 2015

Author's response to reviews:

Reviewer #1:

Thank you very much again for your comment.

Ad 1. We have now stated in the protocol who generated the envelopes used to randomise the patients (page 13, line 11-12).

Reviewer #3:

Thank you again for your comment.

Ad 1. The numerical value of the SD has been added on page 5, line 23.

Ad 2. We have removed [1] from reference 17 on page 26, line 18.